Author's response to reviews

Title: Prevalence of multi-site musculoskeletal symptoms: a French cross-sectional working population-based study

Authors:

Elsa Parot-Schinkel (elparot@chu-angers.fr)
Alexis Descatha (alexis.descatha@inserm.fr)
Catherine Ha (c.ha@invs.sante.fr)
Audrey Petit-Le Manac'h (aupetit@chu-angers.fr)
Annette Leclerc (annette.leclerc@inserm.fr)
Yves Roquelaure (yvroquelaure@chu-angers.fr)

Version: 2 Date: 19 August 2011

Author's response to reviews: see over
Dear Editor,

The modifications realized to answer your request are the following ones:

- The third sentence of the section **Data** was moved in the section **Study design and population** (second sentence);
- And it was so completed: « The Pays de la Loire study was approved by both appropriate national committees: the *Comité consultatif sur le traitement de l’information en matière de recherché dans le domaine de la santé* (CCTIRS n°01-215) and the *Commission nationale de l’informatique et des libertés* (CNIL n°901273). ».

Sincerely Yours

Dr Elsa Parot-Schinkel, M.D.